Author's response to reviews

Title: Germinated brown rice and its bioactives modulate the activity of uterine cells in oophorectomised rats as evidenced by gross cytohistological and immunohistochemical changes

Authors:

Sani M Ismaila (snismaila99@gmail.com)
Maznah Ismail (maznah@medic.upm.edu.my)
Rozi B Mahmud (rozi@medic.upm.edu.my)
Abubakar M Salisu (msabuus@yahoo.com)
MD Zuki Abu Bakar Zakaria (mdzukiabubakar@gmail.com)

Version: 4 Date: 9 July 2013

Author's response to reviews: see over
Dear Editor,

The authors of this manuscript declared no competing interest with PadiBeras Nasional (BERNAS) which will directly or indirectly influence the findings of this study.

Yours Sincerely

Sani IM